\newcommand{\noopsort}[1]{} \newcommand{\printfirst}[2]{#1}
  \newcommand{\singleletter}[1]{#1} \newcommand{\switchargs}[2]{#2#1}
\begin{thebibliography}{10}

\bibitem{andersen91govinfo}
David~F. Andersen and Sharon~S. Dawes.
\newblock {\em {Government Information: A Primer and Casebook}}.
\newblock Prentice-Hall, Englewood Cliffs, NJ, 1991.

\bibitem{bauer03access}
Lujo Bauer.
\newblock {\em {Access Control for the {W}eb via Proof-carrying
  Authorization}}.
\newblock PhD thesis, Princeton University, November 2003.

\bibitem{bauer05distributed}
Lujo Bauer, Scott Garriss, and Michael~K. Reiter.
\newblock {Distributed proving in access-control systems}.
\newblock In {\em Proceedings of the 2005 IEEE Symposium on Security and
  Privacy}, pages 81--95, May 2005.

\bibitem{bernerslee06n3}
Tim Berners-Lee.
\newblock {Notation 3}.
\newblock http://www.w3.org/DesignIssues/Notation3.html, March 2006.

\bibitem{lee01semweb}
Tim Berners-Lee, James Hendler, and Ora Lasilla.
\newblock {The Semantic Web}.
\newblock {\em Scientific American}, May 2001.

\bibitem{bernstein07optarq}
Abraham Bernstein, Christoph Kiefer, and Marcus Stocker.
\newblock {OptARQ: A SPARQL Optimization Approach based on Triple Pattern
  Selectivity Estimation}.
\newblock Technical Report ifi-2007.03, Department of Informatics, University
  of Zurich, March 2007.

\bibitem{bertails08semwebadv}
Alexandre Bertails and Gautier Poupeau.
\newblock {L’apport des technologies du Web sémantique à la gestion des
  données structurées}.
\newblock
  http://www.slideshare.net/lespetitescases/lapport-des-technologies-du-web-sm%
antique-la-gestion-des-donnes-structures, 2008.

\bibitem{rdfsw3c}
Dan Brickley and R.V. Guha.
\newblock {RDF Vocabulary Description Language 1.0: RDF Schema}.
\newblock http://www.w3.org/TR/rdf-schema/, February 2004.

\bibitem{candan96smd}
K.S. Candan, S.~Jajodia, and V.S. Subrahmanian.
\newblock {Secure Mediated Databases}.
\newblock In {\em Proceedings of the Twelfth International Conference on Data
  Engineering}, March 1996.

\bibitem{cherian10sparqlfed}
Mathew Cherian, Lalana Kagal, and Eric Prud'hommeaux.
\newblock {Policy Mediation to Enable Collaborative Use of Sensitive Data}.
\newblock In {\em The Future of the Web for Collaborative Science - World Wide
  Web Conference}, Raleigh, NC, USA, 2010.

\bibitem{Codd70relmodel}
E.F. Codd.
\newblock {A Realtional Model of Data for Large Shared Data Banks}.
\newblock {\em Communications of the ACM}, June 1970.

\bibitem{chester08welcome}
Mitchell D.~Chester Commissioner.
\newblock {Welcome to the Massachusetts Department of Elementary and Secondary
  Education}.
\newblock http://www.doe.mass.edu/mailings/welcome.html, 2008.

\bibitem{bickerton10interview}
Robert Bickerton~(Associate Commissioner) and Robert Lei (Chief Data
  Analyst~MCAS Group).
\newblock {Interview @ Department of Elementary and Secondary Education, 75
  Pleasant St, Malden, MA 02148 on October 13, 2010}.

\bibitem{swig}
{Dave Beazley}.
\newblock {SWIG}.
\newblock http://sourceforge.net/apps/mediawiki/swobjects/index.php, 2010.

\bibitem{dawes96agencyinfoshare}
Sharon~S. Dawes.
\newblock {Interagency Information Sharing: Expected Benefits, Manageable
  Risks}.
\newblock {\em Journal of Policy Analysis and Management}, Summer 1996.

\bibitem{air}
{Decentralized Information Group}.
\newblock {AIR Policy Language}.
\newblock http://dig.csail.mit.edu/2009/AIR/, 2009.

\bibitem{paw}
{Decentralized Information Group and MINDSWAP}.
\newblock {Policy-Aware Web Project}.
\newblock http://www.policyawareweb.org/, 2006.

\bibitem{swobjects}
{Eric Prud'hommeaux}.
\newblock {SWObjects Semantic Web Library}.
\newblock http://sourceforge.net/apps/mediawiki/swobjects/index.php, 2010.

\bibitem{mass10egovben}
Executive~Office for Administration and Finance Mass.Gov.
\newblock {The Benefits of E-Government}, 2010.

\bibitem{dig09secureswfed}
Decentralized~Information Group.
\newblock {Secure Federation Systems for Semantic Web Sources}.
\newblock http://dig.csail.mit.edu/2009/AFOSR/index.html.

\bibitem{dig09dhs}
Decentralized~Information Group.
\newblock {Accountable Information Usage in Fusion Center Information Sharing}.
\newblock http://dig.csail.mit.edu/2009/DHS-fusion/index.html, 2009.

\bibitem{rdfw3c}
RDF~Working Group.
\newblock {Resource Description Framework (RDF)}.
\newblock http://www.w3.org/RDF/, February 2004.

\bibitem{harmon98protpersonal}
J.~Keith Harmon and Rae~N. Cogar.
\newblock {The Protection of Personal Information in Intergovernmental
  Data-Sharing Programs}, 1998.

\bibitem{landsbergen01govinfosystems}
David~Landsbergen Jr. and George~Walken Jr.
\newblock {Realizing the Promise: Government Information Systems and the Fourth
  Generation of Information Technology}.
\newblock {\em Public Administration Review}, March/April 2001.

\bibitem{kagal06self}
Lalana Kagal, Tim Berners-Lee, Dan Connolly, and Daniel Weitzner.
\newblock {Self-Describing Delegation Networks for the Web}.
\newblock In {\em IEEE POLICY '06: Proceedings of the Seventh IEEE
  International Workshop on Policies for Distributed Systems and Networks},
  pages 205--214, Washington, DC, USA, 2006. IEEE Computer Society.

\bibitem{kagal06using}
Lalana Kagal, Tim Berners-Lee, Dan Connolly, and Daniel~J. Weitzner.
\newblock {Using Semantic Web Technologies for Policy Management on the Web}.
\newblock In {\em 21st National Conference on Artificial Intelligence}, 2006.

\bibitem{kagal08using}
Lalana Kagal, Chris Hanson, and Daniel Weitzner.
\newblock {Using Dependency Tracking to Provide Explanations for Policy
  Management}.
\newblock In {\em IEEE Policy 2008}, 2008.

\bibitem{kolas07queryrewrite}
Dave Kolas.
\newblock {Query Rewriting for Semantic Web Information Integration}.
\newblock In {\em Sixth International Workshop on Information Integration on
  the Web}, 2007.

\bibitem{kolovski05towards}
Vladimir Kolovski, Yarden Katz, James Hendler, Daniel Weitzner, and Tim
  Berners-Lee.
\newblock {Towards a policy-aware web}.
\newblock In {\em In Semantic Web and Policy Workshop at the 4th International
  Semantic Web Conference}, 2005.

\bibitem{langegger07distsprql}
Andreas Langegger, Martin Blochl, and Wolfram Woss.
\newblock {Sharing Data on the Grid using Ontologies and distributed SPARQL
  Queries}.
\newblock In {\em 18th International Workshop on Database and Expert Systems
  Applications}, pages 450--454, Washington, DC, USA, 2007. IEEE Computer
  Society.

\bibitem{owlw3c}
Deborah McGuinnes and Frank van Harmelen.
\newblock {OWL Web Ontology Language}.
\newblock http://www.w3.org/TR/owl-features/, February 2004.

\bibitem{mcleod80fedarch}
Dennis McLeod and Dennis Heimbimger.
\newblock {A Federated Architecture for Database Systems}.
\newblock In {\em National Computer Conference}, 1980.

\bibitem{ibm}
N/A.
\newblock None.
\newblock
  http://www.ibm.com/developerworks/data/library/techarticle/dm-0504zikopoulos%
/Part2Figure2.gif, April 2005.

\bibitem{economist10}
N/A.
\newblock {The Data Deluge}.
\newblock {\em The Economist}, February 25 2010.

\bibitem{lod}
N/A.
\newblock {Linked Data - Connect Distributed Data across the Web}.
\newblock http://linkeddata.org/, 2011.

\bibitem{dese10rqmts}
Department of~Elementary and Secondary Education.
\newblock {Massachusetts Comprehensive Assessment System: Participation
  Requirements for Students}.
\newblock http://www.doe.mass.edu/mcas/participation, 2010.

\bibitem{wright10interview}
Sharon Wright (Chief~Information Officer).
\newblock {Interview @ Executive Office of Education, 1 Ashburton Place,
  Boston, MA 02108 on July 22, 2010}.

\bibitem{oregon10egovben}
Oregon.gov.
\newblock {Oregon E-Government Program}.
\newblock http://www.das.state.or.us/DAS/EISPD/EGOV/benefits.shtml, 2010.

\bibitem{sparqlw3c}
Eric Prud'hommeaux and Andy Seaborne.
\newblock {SPARQL Query Language for RDF W3C Recommendation}.
\newblock http://www.w3.org/TR/rdf-sparql-query, January 2008.

\bibitem{quilitz08querying}
Bastian Quilitz and Ulf Leser.
\newblock {Querying Distributed RDF Data Sources with SPARQL}.
\newblock In {\em 5th European Semantic Web Conference (ESWC2008)}, pages
  524--538, June 2008.

\bibitem{sopko07nclbregs}
Kim Sopko and Nancy Reader.
\newblock {Public and Parent Reporting Requirements: NCLB and IDEA
  Regulations}.
\newblock
  http://www.projectforum.org/docs/PublicandParentReportingRequirements-NCLBan%
dIDEARegulations.pdf, 2007.

\bibitem{uriurl}
{Tim Berners-Lee}.
\newblock {Naming and Addressing: URIs, URLs,...}
\newblock http://www.w3.org/Addressing/, 1993.

\bibitem{cwm}
{Tim Berners-Lee and Vladimir Kolovski and Dan Connolly and James Hendler and
  Yoseph Scharf}.
\newblock {A Reasoner for the Web}.
\newblock http://www.w3.org/2000/10/swap/doc/paper/, October 2000.

\bibitem{weng08semrel}
Jinpeng Wang, Zhuang Miao, Yafei Zhang, and Jianjiang Lu.
\newblock {Semantic Integration of relational data using SPARQL}.
\newblock In {\em Second International Symposium on Intelligent Information
  Technology Application}, pages 422--426, 1996.

\bibitem{watugula06thesis}
Sumudu Watugula.
\newblock {Designing Customizable End User Applications using Semantic
  Techologies to Improve Information Management}.
\newblock {MEng Thesis}, MIT, Department of Electrical Engineering and Computer
  Science, May 2006.

\end{thebibliography}

%% This is an example first chapter.  You should put chapter/appendix
%% that you write into a separate file, and add a
%% line \include{yourfilename} to main.tex, where `yourfilename.tex'
%% is the name of the chapter/appendix file.  You can process specific
%% files by typing their names in at the \files= prompt when you run
%% the file main.tex through LaTeX.
\chapter{Introduction}

The amount of digital information in the world has exploded in the
past decade. In 2005, about 150 exabytes of data were created - a
number that has been estimated to have reached 1200 exabytes in
2010~\cite{economist10}. Detecting patterns and extracting useful
information from this deluge of data can significantly benefit firms,
governments, and people by improving efficiencies, and innovating
products and practices, among others. Given the size of this data, it
is necessary to have appropriate tools that can facilitate the
organization and querying of data, which is often contained in
disparate locations. The Semantic Web has the potential to make such
large scale data manipulations possible.

The term Semantic Web encompasses the broad category of technologies
that are aimed at making the World Wide Web more machine-readable.
Some of these technologies include Resource Description Framework
(RDF)~\cite{rdfw3c}, RDF Schema (RDF-S)~\cite{rdfsw3c}, Web Ontology
Language (OWL)~\cite{owlw3c}, the SPARQL Query
Language~\cite{sparqlw3c}. Though the Semantic Web has led to open
data sources that can be queried, extended and re-used
easily~\cite{lod}, on-the-fly integration of data from multiple
heterogeneous and distributed data sources is still a
problem. Furthermore, the ability to perform such seamless
integrations while preserving appropriate security and privacy
policies has not even been addressed.

Several issues need to be addressed before such environments are
developed. For instance, an application developer/user would need to
know what data is present in each source and in which format, so that
appropriate queries may be formulated to obtain the necessary
information. Moreover, if a source's content has been modified, the
application developer needs to adjust the query accordingly to ensure
proper data retrieval.  One possible solution to address these
challenges is the application of the federation paradigm for database
systems~\cite{mcleod80fedarch}.

A federated query involves the concurrent search of multiple
distributed data sources. In a federation model, the access to data
sources is transparent. Under such a system, a user can submit a
single query to the federation engine, which then reformulates the
query to subqueries to suit the various data sources, executes the
subqueries, and returns the merged result. The lack of a shared model
for security and privacy~\cite{dig09secureswfed}, however, impedes the
transparency of the federated model because the federation engine is
unable to execute requests across many secure domains on the fly. This
means that most contemporary federations have static, previously
negotiated policies in place, which prevents dynamic integration of
data.

In this thesis, I present a novel architecture of a data integration
engine that provides secure SPARQL federation. The engine accepts a
single SPARQL query from a client, splits it up into subqueries based
on the information contained in the service descriptions of endpoints
it has in its possession, sends them off to appropriate SPARQL
endpoints, and returns the results to the client. Under the proposed
architecture, clients who use the federation engine are also required
to provide some security credentials, which are used to satisfy the
access policies of those endpoints that contain secure data. Such data
may include personally identifiable information(PII) of individuals
and trade secrets, among others.I also present my implementation of
this architecture, which includes provisions for integration with
Semantic Web-based policy reasoners\cite{cwm, air}, but does not
currently support complete policy enforcement.

\section{Thesis Overview}

The proposed system for SPARQL federation is organized in this thesis
as follows:

Chapter~\ref{ch:background} introduces some of the basic technologies
that makes query federation possible in a Semantic Web environment. A
basic overview of federated querying is provided and the differences
between relational and RDF databases are illustrated. Then, some of
the recent advances in federation functionality in Semantic Web
technologies is explored.

Chapter~\ref{ch:usecase} describes the application of a SPARQL
federation system in the context of government data sharing. When
organizations see the business case for expressing their data in
formats that suit the Semantic Web, they would begin developing data
systems that facilitate enterprise level data sharing. The presence of
such data sources would accelerate the spread of Semantic Web
Technologies on a truly Web scale. This detailed use case highlights
the many advantages of adopting SPARQL federation systems for the
Massachusetts Department of Education over traditional databases.

Chapter~\ref{ch:architecture} discusses the overall architecture of
the system and the design choices that were made during its
development. The various components of the system are described and
their features and limitations are identified.

Chapter~\ref{ch:implementation} details the implementation
of the design. The various software tools that made the design
possible are identified and their relevance is described. 

Chapter~\ref{ch:results} discusses the performance of the system
when presented with endpoints and queries with different
characteristics. The methodology for the selection of queries is
presented and the relevant metrics are illustrated.

Chapter~\ref{ch:summary} provides a summary of the contributions of
this work. The limitations of the system are discussed and future
areas for expanding on the work are identified.
